Author's response to reviews

Title: Unusual Causes of Parkinsonism: a case series of potential cause-effect relationship

Authors:

Solomon O Ugoya (doc souls@yahoo.com)
Emmanuel I Agaba (eiagaba@yahoo.com)
Comfort A Daniyam (comfort_13@yahoo.com)

Version: 6 Date: 5 November 2010

Author's response to reviews: see over
The Editor,

JMCR

Attention: Editorial Team

RE: MS: 6129007063691597- Submission of Manuscript titled: Unusual causes of parkinsonism: a case series

We are re-submitting the corrected version (galley proof) of our paper for possible publication in your reputable journal.

The Abstract in the manuscript matches exactly with the Abstract in the submission system we ensured the patient identification (age, gender and ethnicity) is stated consistently in the manuscript and the abstract. Lady' was changed to 'female'/woman'. In both the text and abstract.

References: references 14, 15,16 and 17 are cited in the consecutive order and corrected to correspond with those in the text.

We moved disclosure from the title page. We ensured the Competing interests section precedes the Consent section.

Yours sincerely,

Dr. Solomon O. Ugoya.